Author's response to reviews

Title: Simultaneous Exposure to Multiple Heavy Metals and Glyphosate May Contribute to Sri Lankan Agricultural Nephropathy

Authors:

Channa Jayasumana (jayasumanalk@yahoo.com)
Sarat Gunatilake (sarathg@csulb.edu)
Sisira Sirbaddana (sisira.sirbaddana@gmail.com)

Version: 3 Date: 7 April 2015

Author's response to reviews: see over
Dear Editor,

Thank you very much for your kind consideration. Herewith we submit the revised manuscript. Following changes were made to the manuscript as requested by editorial team.

1) Title page – E mail address of all authors and the full postal address of the submitting author were included.
2) Introduction section is renamed as Background.
3) Authors' contributions section is included before the Acknowledgements and Reference list.

Suggesting reviewers

1) Dr.Y.J. Anupama ,Nephrology unit, Nanjappa Hospital, Shimoga 577201 ,Karnataka, India, +919686213290, anupamayj@gmail.com
2) Dr.Pedro Ordunez, Pan American Health Organization, Washington, D.C.,United States of America, ordunezp@paho.org
3) Dr.Stephanie Seneff, Massachusetts Institute of Technology, Rm G-438 MIT Stata Center, 32 Vassar Street, Cambridge, MA 02139, USA. TEL.: +1-617-253-0451. seneff@csail.mit.edu

Thanking you,

Dr.Channa Jayasumana
Department of Pharmacology,
Faculty of Medicine,
Rajarata University of Sri Lanka,
Anuradhapura, Sri Lanka.
+94 714 39 39 89